Author's response to reviews

Title: The Box-Cox power transformation on nursing sensitive indicators: Does it matter if structural effects are omitted during the estimation of the transformation parameter?

Authors:

Qingjiang Hou (qhou@kumc.edu)
Mahnken D Jonathan (jmahnken@kumc.edu)
Gajewski J Byron (bgajewski@kumc.edu)
Nancy Dunton (ndunton@kumc.edu)

Version: 5 Date: 18 August 2011

Author's response to reviews: see over
Formatting changes to be made:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Checked.

Tables: please provide titles for your tables. These should be placed above the table.

We added a title for each table.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

We retyped all three tables.